

July 6, 2015

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 18914-Revised manuscript).

Title: Research progress and prospects of markers for liver cancer stem cells

Author: Cheng-Pei Zhu, An-Qiang Wang, Hao-Hai Zhang, Xue-Shuai Wan, Xiao-Bo Yang, Shu-Guang Chen, Hai-Tao Zhao

Name of Journal: *World Journal of Gastroenterology*

ESPS Manuscript NO: 18914

The manuscript has been improved according to the suggestions of reviewers and editor:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer and editor

(1) Correct a small amount of language errors

(2) Three reviewers evaluate the topic theme of article in general, without any other suggestions

(3) Add core tip in the text

(4) Provide a language editing certificate

(5) All changes are marked in red font

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Gastroenterology*.

Sincerely yours,

A handwritten signature in black ink on a light-colored rectangular background. The signature reads "Haitao Zhao" in a cursive, slightly slanted script.

Haitao Zhao, MD
Department of Liver Surgery
Peking Union Medical College Hospital
Chinese Academy of Medical Sciences
1 Shuaifuyuan, Wangfujing, Beijing
Fax: +86-10-69156042
E-mail:zhaht@pumch.cn